Author's response to reviews

Title: Effectiveness of Evidence-Based Medicine Training for Undergraduate Students in a Chinese Military Medical University: a self-controlled trial

Authors:

xiangyu ma (xymacq@hotmail.com)
bin xu (191436213@qq.com)
qingyun liu (928365207@qq.com)
hongyan xiong (hongyanxiong@sohu.com)
yafei li (xymacq@gmail.com)

Version: 3 Date: 18 June 2014

Author's response to reviews:

Authors' Responses to Reviewers' Comments

RE: Effectiveness of Evidence-Based Medicine Training for Undergraduate Students in a Chinese Military Medical University: a self-controlled trial (1030815228120956)

Dear Mr Aldrin Ulep,

Thank you very much for your letter, and to the reviewers for their thoughtful critique of our manuscript. Specific responses to each comment follow, with our responses being highlighted with RED color.

1. Ethics approval
Please add the full name and affiliation of the ethics committee that approved the study to the Methods section.
Response: Added.
The strategy was approved by the ethics committee of Third Military Medical University.

2. Consent
Please add a formal consent to participate statement to the manuscript, explicitly stating whether consent was written or verbal.
Response: Added.
a verbal consent was obtained from all participants (Appendix 2).

Best Regards
Yafei Li, M.D., MPH, Ph.D